Author's response to reviews

Title: Influence of tidal volume on pulse pressure variation and stroke volume variation during experimental intra-abdominal hypertension.

Authors:

Franco Diaz (francodiazr@gmail.com)
Benjamin Erranz (berranz@gmail.com)
Alejandro Donoso (adonosof@gmail.com)
Tatiana Salomon (tattysalomon@hotmail.com)
Pablo Cruces (pcrucesr@gmail.com)

Version: 4 Date: 9 August 2015

Author's response to reviews: see over
August 8th, 2015

Prof Marcelo Gama de Abreu
Associate Editor, BMC Anesthesiology Journal

Dear Prof. Gama de Abreu,

Thank you for the quick review and process of the resubmission of our Manuscript (ID 447792983161274) entitled "Influence of tidal volume on pulse pressure variation and stroke volume variation during experimental intra-abdominal hypertension". We made all the editions requested by the editorial office as listed below.

1. Manuscript was reformatted, we added a Conclusions sections and deleted the section key messages.
2. Additional files were reformatted according to the instructions for authors.
3. Additional files were renamed accordingly, and Figure S1B was replaced by Additional file 3.
4. We added the section “Additional files” at the end of the manuscript, listing the name, format and description of the additional files.

Thank you for the opportunity to revise our manuscript. We feel that it has been strengthened considerably as a result of the suggested edits.

Sincerely

Franco Díaz, M.D.
Department of Pediatrics
Division of Pediatric Critical Care.
University of Alabama at Birmingham.

Benjamín Erranz, PhD.
Centro de Medicina Regenerativa, Facultad de Medicina
Clínica Alemana-Universidad del Desarrollo.
Santiago, Chile

Pablo Cruces, M.D.
Centro de Investigación de Medicina Veterinaria, Universidad Andres Bello.
Pediatric Intensive Critical Unit, Hospital El Carmen de Maipú.
Santiago, Chile